Author's response to reviews

Title: Prevalence, impacts and medical managements of premenstrual syndrome among female students: cross-sectional study in college of health science, Mekelle university, Mekelle, northern Ethiopia

Authors:

Fikru W Tolossa (fikrewaf@gmail.com)
Mebratu L Bekele (mebleq@gmail.com)

Version: 3 Date: 16 October 2013

Author's response to reviews: see over
Date: 16/10/2013

Subject: SUBMISSION OF A MANUSCRIPT

Dear Editor,

We are enclosing herewith a manuscript entitled “prevalence, impacts and medical managements of premenstrual syndrome among female students: cross-sectional study in college of health science, Mekelle university, Mekelle, northern Ethiopia” for publication in “BMC Women's Health” for possible evaluation.

We recognized that “BMC Women's Health” is most appropriate journal for researches related with women’s behavioral, mental and emotional health including issues related with menstrual cycle and disorders associated with it. With the submission of this manuscript, we would like to undertake that the above mentioned manuscript has not been published elsewhere, accepted for publication elsewhere or under editorial review for publication elsewhere.

The submitted manuscript is a Research Article, which was conducted in the months March and April, 2013. This manuscript is a reality of all the authors who contributed to the publication. There is no Conflict of interest in this work.

- Number of words and characters (without spaces) of the topic 23 (168)
- Number of words of manuscript 4,080
- Number of words of abstract 333
- Number of references 36

Best Regards

Mebratu Legesse Bekele (Corresponding author)
College of Health Sciences and Medicine, Wolaita Sodo University,
P.O.Box: 138;
Tel: +251-926-105896;
Email: mebleg@gmail.com

Wolaita Sodo, Ethiopia.